Author's response to reviews

Title: Successful transureteropyeloneostomy after heminephrectomy of a bilateral-hydronephrotic horseshoe kidney

Authors:

Holger Gerullis (holger.gerullis@gmx.net)
Christoph Eimer (christoph.eimer@web.de)
Dietmar Betz (dietmarbetz@hotmail.com)
Thomas Otto (thomas_otto@lukasneuss.de)

Version: 5 Date: 30 April 2008

Author's response to reviews: see over
Cover letter

Dear Dr. Agarwal

Thank You for the second revision of our manuscript. The article now underwent another rigorous evaluation regarding orthography and grammar.

Since the other two reviewers were totally satisfied with the implemented modifications regarding the initially criticised figures we would like to maintain the line diagrams. Another change, withdrawing the newly implemented line diagrams, would probably have provoked discrepancies among reviewer one and two who claimed a graphic explanation for the conducted surgery in order to plasticize the surgical approach and the pre- and post operative settings.

Thus we decided to keep the figures in the present form.

H. Gerullis